Author’s response to reviews

Title: The General Health Questionnaire-28 (GHQ-28) as an outcome measurement in a randomized controlled trial in a Norwegian stroke population

Authors:

Ellen Gabrielsen Hjelle (e.g.hjelle@medisin.uio.no)
Manuela Zucknick (manuela.zucknick@medisin.uio.no)
Line Bragstad (l.k.bragstad@medisin.uio.no)
Marit Kirkevold (marit.kirkevold@medisin.uio.no)
Bente Thommessen (bthommessen@yahoo.no)
Unni Sveen (unnsve@oslomet.no)

Version: 1 Date: 28 Oct 2018

Author’s response to reviews:

Cover letter with point to point response to the comments has been uploaded as supplementary material